Author's response to reviews

Title: Cost-effectiveness analysis of PCR for the rapid diagnosis of pulmonary tuberculosis

Authors:

Luciene Cardoso Scherer (luciene.scherer@hotmail.com)
Rosa Dea Sperhacke (deasperhacke@hotmail.com)
Antonio Ruffino-Netto (aruffino@fmrp.usp.br)
Maria Lúcia Rosa Rossetti (rossetti@terra.com.br)
Claudia Vater (mcvater@nesc.ufrj.br)
Paul Klatser (paul.klatser@falw.vu.nl)
Afrânio Lineu Kritski (kritski@gmail.com)

Version: 8 Date: 22 December 2009

Author's response to reviews: see over
Rio de Janeiro, 22 December 2009

Andrea Bucceri PhD
Assistant Editor
BMC-series journals

Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com

R4

This is a covering letter with a point-by-point description of the changes made in manuscript

Cost-effectiveness analysis of PCR for the rapid diagnosis of pulmonary tuberculosis MS: 9534281932436370

Manuscript revised with point-by-point description of the changes made.

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is

¿First name Initial Last name¿ (e.g. Joe F Bloggs)

The manuscript was changed in the new version.
References: References should be cited in the text using consecutive numbers in square brackets. Reference 33 appears before reference 32 - please switch these references, or renumber the reference list accordingly.

The manuscript was changed in the references as the reviewer indicated (new version).

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbers: Please remove these

The manuscript was changed in the references as the reviewer indicated (new version).

Highlighting/tracking: Please remove all highlighting, coloured text and tracking from the manuscript.

The manuscript was changed in the references as the reviewer indicated (new version).

Abbreviations: Please format these in to a paragraph.

The manuscript was changed in the references as the reviewer indicated (new version).
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript was changed in the references as the reviewer indicated (new version).

Luciene Cardoso Scherer M.D.
Center of Improvement Scientific and Technologic –CDCT
State Foundation by Research in Health –FEPPS/RS
Av Ipiranga 5400, 3° andar
Porto Alegre, Brazil
CEP 90610-000.
phone/fax: (+55) 51 3352 0336
e-mail:luciene.scherer@hotmail.com

Afranio Kritski M.D., PhD.
Tuberculosis Academic Program, Coordinator
IDT-HUCFF Hospital Complex
Internal Medicine Department, Chief
Medical School of Federal University of Rio de Janeiro
Av Brigadeiro Trompowsky s/n - Ilha Fundao - Predio HUCFF 4 andar
Rio de Janeiro, Brazil

CEP 21 941 590/ phone/fax: 55 21 2550 69 03/ email: kritskia@gmail.com